Author's response to reviews

Title: Cost-effectiveness of a structured progressive task-oriented circuit class training programme to enhance walking competency after stroke: protocol of the FIT-STROKE Trial

Authors:

Ingrid GL van de Port (i.v.d.port@dehoogstraat.nl)
Lotte Wevers (l.wevers@dehoogstraat.nl)
Hanneke Roelse (h.roelse@dehoogstraat.nl)
Lenneke van Kats (l.v.kats@dehoogstraat.nl)
Eline Lindeman (e.lindeman@dehoogstraat.nl)
Gert Kwakkel (g.kwakkel@planet.nl)

Version: 2 Date: 30 July 2009

Author's response to reviews:

Revisions have been made as required in the email of July 29th 2009
- References have been updated
- Figure has been cited in the text
- Citation of the references has been changed
- Methods and Design has been renamed